

 British Society for Rheumatology
Rheumatology
Rheumatology

Clinical science

Real-world evidence for secukinumab in UK patients with psoriatic arthritis or radiographic axial spondyloarthritis: interim 2-year analysis from SERENA

Karl Gaffney (1)^{1,*}, Nicola Gullick², Kirsten MacKay³, Yusuf Patel⁴, Raj Sengupta (1)⁵, Tom Sheeran⁶, Louise Hemmings⁷, Paula Pamies⁷

¹Department of Rheumatology, Norfolk and Norwich University Hospitals, NHS Foundation Trust, Norwich, UK ²University Hospital Coventry & Warwickshire, Warwick Medical School, University of Warwick, Coventry, UK

³Rheumatology, Torbay and South Devon NHS Foundation Trust, Torquay, UK

⁴Hull University Teaching Hospitals NHS Trust, Hull, UK

⁵Royal National Hospital for Rheumatic Diseases, Royal United Hospitals, Bath, UK

⁶University of Wolverhampton, Royal Wolverhampton NHS Trust, Wolverhampton, UK

⁷Immunology, Novartis Pharmaceuticals UK Ltd, London, UK

*Correspondence to: Karl Gaffney, Department of Rheumatology, Norfolk and Norwich University Hospitals, NHS Foundation Trust, Colney Ln, Norwich NR4 7UY, UK. E-mail: karl.gaffney@nnuh.nhs.uk

Abstract

Objectives: The aim was to evaluate retention rates for secukinumab in patients with active PsA or radiographic axial spondyloarthritis (r-axSpA) treated in routine UK clinical practice.

Methods: SERENA (CAIN457A3403) is an ongoing, non-interventional, international study of patients with moderate-to-severe chronic plaque psoriasis, active PsA or active r-axSpA, who had received secukinumab for \geq 16 weeks before enrolment. The primary objective of this interim analysis was to assess treatment retention rates in patients with PsA or r-axSpA who were enrolled and followed for \geq 2 years at centres in the UK. The safety analysis set includes all patients who received at least one dose of secukinumab. The target population set includes all patients who fulfilled the patient selection criteria.

Results: The safety set comprised 189 patients (PsA, n=81; r-axSpA, n=108), and the target population set comprised 183 patients (PsA, n=78; r-axSpA, n=105). In the safety set, 107 patients (45 of 81 with PsA and 62 of 108 with r-axSpA) had previously received a biologic agent. Retention rates were similar between patients with PsA and r-axSpA after 1 year (PsA 91.0%, 95% Cl: 84.0, 98.0; r-axSpA 89.2%, 95% Cl: 82.7, 95.7) and 2 years (PsA 77.6%, 95% Cl: 67.6, 87.7; r-axSpA 76.2%, 95% Cl: 67.4, 85.0) of observation. Overall, 17.5% of patients (33 of 189) experienced at least one treatment-related adverse event, and 12.7% of patients (24 of 189) discontinued secukinumab because of adverse events.

Conclusion: This analysis of real-world data from the UK demonstrates high retention rates for secukinumab over 2 years in patients with PsA or r-axSpA, with a favourable safety profile.

Lay Summary

What does this mean for patients?

Secukinumab is a medication that can be used to treat patients with severe psoriatic arthritis (PsA) or axial spondyloarthritis (axSpA). Some patients may ultimately stop taking secukinumab for many potential reasons, however there is little information published on how many actually do this. SERENA is a study examining secukinumab in patients with PsA or axSpA. This article describes the portion of the SERENA study that looked specifically at patients treated in the UK. Patients who had taken secukinumab for at least 16 weeks before joining the SERENA study were assessed over at least 2 years. In total, 194 patients (82 with PsA and 112 with axSpA) took part at UK centres. Secukinumab was still being taken by 90% of patients after 1 year and 77% after 2 years. In most patients, secukinumab continued to reduce pain and swelling over the 2 years of observation. Overall, 17.5% of patients had at least one side effect, and 12.7% of patients, some taking secukinumab because of side effects. The UK results show that continuing to take secukinumab leads to sustained benefits for patients, and is well tolerated.

Keywords: drug survival, PsA, radiographic axial spondyloarthritis, real-world evidence, secukinumab

Key messages

- Few data exist regarding retention of secukinumab in UK patients with PsA/axial spondyloarthritis.
- In UK patients, retention rates were 90% and 77% after 1 and 2 years, respectively.
- The results demonstrate sustained effectiveness, good retention rates and a favourable safety profile for secukinumab.

© The Author(s) 2023. Published by Oxford University Press on behalf of the British Society for Rheumatology.

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

Introduction

PsA and axial spondyloarthritis (axSpA) are chronic autoimmune disorders that share common immunopathological and clinical features [1, 2]. PsA is characterized by inflammation of the axial and peripheral joints, entheses and skin [1, 2]. Extra-musculoskeletal manifestations can include uveitis and IBD [1]. PsA is associated with impaired physical function and poor quality of life (QoL) [3–5]. In the UK, the prevalence of PsA is estimated to be 0.19% [6].

AxSpA is characterized by involvement of the axial skeleton and SI joints [1, 2]. Changes in the SI joints can be detected by conventional radiography in patients with radiographic axSpA (r-axSpA), or by MRI alone in patients with non-radiographic axSpA (nr-axSpA) [1]. Clinical hallmarks include inflammatory back pain, enthesitis, stiffness and fatigue, which lead to functional impairment and reduced QoL [1, 3, 7]. Approximately one in three patients with axSpA have an extra-musculoskeletal manifestation [8]. The prevalence of axSpA in the UK is estimated to be 0.3% [9].

IL-17-mediated immune responses are important in the pathogenesis of PsA and axSpA, leading to joint inflammation, joint destruction and new bone formation [1, 2, 10, 11]. Secukinumab is a fully human anti-IL-17A mAb approved for the treatment of moderate-to-severe plaque psoriasis in patients aged ≥ 6 years, in adults with active PsA and an inadequate response to prior DMARD therapy, in adults with r-axSpA who have responded inadequately to conventional therapy, and in adults with nr-axSpA who have responded inadequately to NSAIDs and who have objective evidence of inflammation (i.e. via MRI or elevated CRP levels) [12]. The clinical trial programme for secukinumab has demonstrated sustained efficacy and a consistent safety profile in patients with psoriasis [13], PsA [14–18], r-axSpA [19–22] and nr-axSpA [23].

Evidence is now accruing on secukinumab for the treatment of PsA or r-axSpA in routine clinical practice [24–31]; however, little real-world evidence exists regarding the long-term retention of secukinumab in patients with PsA and r-axSpA in the UK. Therefore, we conducted an interim analysis of data from an ongoing real-world study to evaluate the retention, safety and effectiveness of secukinumab in patients with PsA or r-axSpA in the UK.

Methods

SERENA (CAIN457A3403) is a non-interventional study in patients with moderate-to-severe chronic plaque psoriasis, active PsA or active r-axSpA, and is currently ongoing in 19 countries (Austria, Belgium, Bulgaria, Croatia, Czech Republic, Estonia, France, Germany, Greece, Hungary, Ireland, Israel, Italy, the Netherlands, Portugal, Russia, Slovenia, Switzerland and the UK) [32]. The study design, patient selection criteria and a comprehensive description of the outcome measures and data analysis plan have been published elsewhere [33]. All included patients provided written informed consent. This study was conducted in accordance with Good Clinical Practice and the Declaration of Helsinki of 1964 and its later amendments. The study was approved by the East of England Ethics committee. Eligible patients had active psoriasis, PsA or r-axSpA and were prescribed secukinumab by the treating physician. All patients should have received secukinumab for ≥ 16 weeks before enrolment [32].

Patients are to be observed for a minimum of 5 years or until secukinumab is discontinued.

Data collected include demographics, prior treatment for psoriasis, PsA or r-axSpA, and disease-specific effectiveness and OoL measures [32]. For patients with PsA, assessments include the 78 Tender Joint Count (TJC), 76 Swollen Joint Count (SJC) and dactylitis [34, 35], Leeds Enthesitis Index [36], Physician Global Assessment (PGA) [35], pain as measured by a 10 cm visual analogue scale (VAS) [37], HAQ-Disability Index (HAQ-DI) and Functional Assessment of Chronic Illness Therapy-Fatigue (FACIT-Fatigue) [35]. For patients with r-axSpA, assessments include BASDAI [38], CRP or high-sensitivity CRP (hsCRP) [39], Ankylosing Spondylitis DAS (ASDAS) [38], total spinal pain as measured by VAS (0-10 cm) and patient's global assessment of disease activity [40]. Safety was assessed by reports of treatment-emergent adverse events (TEAEs), serious adverse events (SAEs) and adverse events (AEs) of special interest, defined as major adverse cardiac events (MACEs), Candida infections, IBD, uveitis and malignancy.

The primary objective of SERENA is to document the longterm retention of secukinumab in routine clinical practice [32]. Eligible patients were aged ≥ 18 years, with moderate-tosevere plaque psoriasis, PsA or AS, and had received secukinumab for ≥ 16 weeks before study enrolment. Patients with medical or psychological conditions that might prevent them from participating in the study and those participating in other clinical trials were excluded.

The safety analysis set includes all patients who received at least one dose of secukinumab after providing written informed consent, and the target population set includes all patients who fulfilled the inclusion/exclusion criteria. The analysis reported herein is restricted to the subset of patients with PsA or r-axSpA who were enrolled and followed for ≥ 2 years at centres in the UK. At the time of trial initiation, secukinumab was not approved in the UK for the treatment of nr-axSpA; therefore, all axSpA patients had r-axSpA.

Results

In total, 194 patients (82 with PsA and 112 with r-axSpA) were enrolled at UK centres between 26 July 2017 and 12 October 2018. Five patients (one with PsA and four with r-axSpA) were excluded from the safety set because they did not receive secukinumab after providing informed consent. Thus, the safety set comprised 189 patients (81 with PsA and 108 with r-axSpA). In addition to the five individuals excluded from the safety set, six patients were excluded from the target analysis population (two patients with PsA who did not fulfil the patient selection criteria; one patient with PsA and two patients with r-axSpA who had protocol deviations; and one patient with r-axSpA who did not fulfil the selection criteria and had a protocol deviation). Thus, the target population set comprised 183 patients (78 with PsA and 105 with r-axSpA). At the time of the interim analysis, the mean age was 49 years, 61.4% of the target population were male, and the mean time since diagnosis of PsA or r-axSpA was 9.1 years and 11.2 years, respectively (Table 1).

As per the protocol, all patients had been receiving secukinumab at the time of enrolment (baseline). The mean duration of treatment with secukinumab was 0.5 years in patients with PsA and 0.7 years in patients with r-axSpA (Table 1). At baseline, for patients with PsA, the mean (s.D.) SJC was 4.5 (6.0), the mean TJC was 12.8 (16.0), the prevalence of dactylitis

Table 1. Patient characteristics at the time of enrolment (safety set)

Variable	PsA	r-axSpA	Total
	(n = 81)	(n = 108)	(n = 189)
Age, mean (s.D.), years ^a	50.4 (10.9)	47.9 (12.4)	49.0 (11.8)
Male, <i>n</i> (%)	43 (53.1)	73 (67.6)	116 (61.4)
White, <i>n</i> (%)	72 (88.9)	95 (88.0)	167 (88.4)
BMI, mean (s.D.), kg/m ²	30.7 (5.7)	28.2 (5.9)	29.0 (5.9)
<18.5, n (%)	0	1 (1.5)	1 (1.0)
>18.5 to <25, n (%)	4 (12.1)	19 (27.9)	23 (22.8)
>25 to $<30, n$ (%)	10 (30.3)	21 (30.9)	31 (30.7)
>30, n(%)	19 (57.6)	27 (39.7)	46 (45.5)
Smoking status, $n(\%)^{b}$			
Current	15 (18.8)	25 (23.1)	40 (21.3)
Former	23 (28.8)	28 (25.9)	51 (27.1)
Never	28 (35.0)	39 (36.1)	67 (35.6)
Unknown	14 (17.5)	16 (14.8)	30 (16.0)
Time since disease diagnosis, mean	9.1 (7.8)	11.2 (11.2)	_
(s.D.), years			
Duration of secukinumab treat-	0.5(0.3)	0.7 (0.4)	0.6 (0.3)
ment prior to inclusion, mean	[0.2, 1.4]	[0.1, 2.0]	[0.1, 2.0]
(s.D.) [range], years		- / -	- / -

^a For patients whose age was missing and only year of birth collected, age was calculated as 1 July in the year of birth.

⁶ Proportions do not add up to 100% because data were missing for some patients.

r-axSpA: radiographic axial spondyloarthritis.

was 11.9%, and the prevalence of enthesitis was 63.0%. At baseline, for patients with r-axSpA, the mean BASDAI score was 3.4 (2.7), mean VAS scores for nocturnal back pain and total back pain were 3.71 (2.94) and 3.52 (2.82), respectively, and the mean ASDAS-CRP score was 2.4 (1.1). At the time of enrolment, 2.6% of patients (5 of 189) had co-morbid uveitis (one patient with PsA and four patients with r-axSpA) and 1% of patients (2 of 189) had co-morbid IBD (one patient with PsA had Crohn's disease and one patient with r-axSpA had ulcerative colitis).

Treatment history and concomitant treatment

Of the 189 patients in the safety set, 82 (43.4%) were biologic naïve and 107 (56.6%) had received previous treatment with a biologic agent before initiation of secukinumab treatment (Table 2). Among patients with PsA, 55.6% (45 of 81) had previously received at least one biologic agent, 19.8% (16 of 81) had received two, and 18.5% (15 of 81) had received three or more before starting secukinumab. Of the 45 individuals who had received a biologic agent, 40 (88.9%) had discontinued because of lack of efficacy.

Among patients with r-axSpA, 57.4% (62 of 108) had previously received at least one biologic agent, 15.7% (17 of 108) had received two agents, and 11.1% (12 of 108) had received three or more agents before starting secukinumab therapy. Of the 62 individuals who had received a biologic, 52 (83.9%) had discontinued biologics owing to lack of efficacy.

Of the patients with PsA who were receiving secukinumab, 35.8% (29 of 81) were receiving a concomitant NSAID and 63.0% (51 of 81) were receiving a concomitant DMARD alone or in combination with other agents (Table 2). In patients with r-axSpA who were receiving secukinumab, 46.3% (50 of 108) were receiving a concomitant NSAID and 21.3% (23 of 108) were receiving a concomitant DMARD alone or in combination with other agents (Table 2). A substantial minority of patients [21.0% (17 of 81) with PsA and

Table 2. Treatment received before and concomitant with secukinumab (safety set)

Treatment, n (%)	PsA (n = 81)	r-axSpA (<i>n</i> = 108)	Total (<i>n</i> = 189)
Prior treatment history with a			
biologic ^{a,b}			
Received ≥ 1 prior biologic	45 (55.6)	62 (57.4)	107 (56.6)
Received 1 prior biologic	14 (17.3)	33 (30.6)	47 (24.9)
Received 2 prior biologics	16 (19.8)	17 (15.7)	33 (17.5)
Received ≥ 3 prior biologics	15 (18.5)	12 (11.1)	27 (14.3)
Treatment regimens before			
secukinumab			
NSAID	5 (6.2)	17 (15.7)	22 (11.6)
DMARD	16 (19.8)	6 (5.6)	22 (11.6)
NSAID plus DMARD	6 (7.4)	1 (0.9)	7 (3.7)
Biologic (or biosimilar)	13 (16.0)	25 (23.1)	38 (20.1)
NSAID plus biologic (or	3 (3.7)	21 (19.4)	24 (12.7)
biosimilar)			
DMARD plus biologic (or	17 (21.0)	8 (7.4)	25 (13.2)
biosimilar)			
NSAID, DMARD plus biologic	12 (14.8)	8 (7.4)	20 (10.6)
(or biosimilar)			
Not documented/other	9 (11.1)	22 (20.4)	31 (16.4)
Treatment concomitant with			
secukinumab ^{c,d}			
NSAID	13 (16.0)	42 (38.9)	55 (29.1)
DMARD	24 (29.6)	12 (11.1)	36 (19.0)
NSAID plus DMARD	16 (19.8)	6 (5.6)	22 (11.6)
Biologic (or biosimilar)	1 (1.2)	1 (0.9)	2(1.1)
NSAID plus biologic (or	0	2 (1.9)	2(1.1)
biosimilar)			
DMARD plus biologic (or	0	1 (0.9)	1(0.5)
biosimilar)			
Other	10 (12.3)	6 (5.6)	16 (8.5)
No documented treatment	17 (21.0)	38 (35.2)	55 (29.1)

^a Among patients with PsA, prior biologic treatments included adalimumab (n = 31, 38.3%), etanercept (n = 25, 30.9%), golimumab (n = 16, 19.8%), ustekinumab (n = 15, 18.5%), infliximab (n = 6, 7.4%), certolizumab pegol (n = 5, 6.2%) and other biologic (n = 3, 3.7%).

^b Among patients with AS, prior biologic treatments included adalimumab (n = 37, 34.3%), etanercept (n = 31, 28.7%), golimumab (n = 10, 9.3%), certolizumab pegol (n = 10, 9.3%), infliximab (n = 2, 1.9%) and other biologic (n = 4, 3.7%).

^c Among patients with PsA, concomitant DMARD therapy included MTX (n = 28, 34.6%), SSZ (n = 10, 12.3%), LEF (n = 6, 7.4%), HCQ (n = 1, 1.2%) and CSA (n = 1, 1.2%).

^d Among patients with r-axSpA, concomitant DMARD therapy included MTX (n = 11, 10.2%), SSZ (n = 9, 8.3%) and LEF (n = 2, 1.9%). r-axSpA: radiographic axial spondyloarthritis.

35.2% (38 of 108) with r-axSpA] were receiving no other documented treatment in combination with secukinumab.

Main outcomes (target population set)

The recommended dose of secukinumab for patients with PsA and axSpA is 150 mg (300 mg in patients with PsA and concomitant moderate-to-severe plaque psoriasis or an inadequate response to a TNF- α inhibitor), with the possibility of increasing to 300 mg as needed. At baseline, most patients were receiving secukinumab at a dose of 150 mg (74.0%, 134 of 181); however, the modal dose at baseline differed by indication. In the subset of patients with PsA, more patients were receiving secukinumab 300 mg (56.6%, 43 of 76; dosing interval of 4 weeks) at baseline than 150 mg (baseline dose was missing for two patients), which is likely to reflect the proportion of patients with prior inadequate response to a TNF- α inhibitor. In contrast, in the subset of patients with r-axSpA, most patients (96.2%, 101 of 105) were receiving 150 mg at baseline, whereas only 3.8% (4 of 105) were receiving the 300 mg dose.

At the time of the interim analysis, secukinumab treatment was ongoing in 73.2% of patients (134 of 183) overall in the target analysis population; 78.2% of patients (61 of 78) with PsA and 69.5% of patients (73 of 105) with r-axSpA. Of the 17 patients with PsA who discontinued secukinumab, 9, 3 and 2 individuals, respectively, had discontinued treatment for lack of efficacy, physician decision and patient decision, while 3 patients discontinued for other reasons (Supplementary Table S1, available at *Rheumatology Advances in Practice* online). Likewise, of the 32 patients with r-axSpA who discontinued secukinumab, 11, 8 and 6 had discontinued treatment for the same three reasons, respectively, while 7 patients discontinued treatment for other reasons (Supplementary Table S1, available at *Rheumatology Advances in Practice* online).

The mean time to treatment discontinuation was similar between patients with PsA and r-axSpA (Fig. 1). High retention rates were observed after 1 year (90.0%, 95% CI: 85.3, 94.7) and 2 years (76.8%, 95% CI: 70.3, 83.3) of treatment with secukinumab. Moreover, retention rates were similar in patients with PsA and r-axSpA after 1 year of observation (PsA 91.0%, 95% CI: 84.0, 0.98; r-axSpA 89.2%, 95% CI: 82.7, 95.7) and after 2 years of observation (PsA 77.6%, 95% CI: 67.6, 87.7; r-axSpA 76.2%, 95% CI: 67.4, 85.0).

When patients were analysed according to their biologic treatment history, the mean time to discontinuation was



Figure 1. Time to treatment discontinuation. r-axSpA: radiographic axial spondyloarthritis

similar between biologic-naïve and biologic pre-treated patients (Fig. 2). However, fewer treatment-naïve r-axSpA patients than treatment-naïve PsA patients discontinued treatment after 2 years of observation.

Treatment interruptions of >2 months were reported in 16 patients (5 of 78 with PsA and 11 of 105 with r-axSpA) after a mean (s.d.) treatment duration of 536 (301) days in patients with PsA and 408 (323) days in patients with r-axSpA. Reasons for treatment interruptions among patients with PsA included AEs (n=3) and missing/other reasons (n=2), and in patients with r-axSpA they included coronavirus disease 2019-related reasons (n=2), patient decision (n=1), treatment change according to product information (n=1) and missing/other reasons (n=7). The mean (s.d.) duration of these treatment interruptions was 157 (91) days in patients with PsA and 229 (226) days in patients with r-axSpA.

In patients with PsA, mean SJCs and TJCs remained stable throughout 2 years of treatment with secukinumab (Table 3). Overall, few patients had dactylitis or enthesitis, and the absolute number decreased over 2 years of treatment. Other efficacy outcomes remained stable throughout 2 years of observation, including total pain scores, the proportion of patients rated as 0/1 on the PGA, the proportion of patients with nail involvement and QoL measures (mean HAQ-DI and FACIT-Fatigue scores).

In patients with r-axSpA, mean BASDAI scores were sustained, and scores for total back pain and nocturnal back pain remained stable throughout 2 years of treatment with secukinumab (Table 3). Likewise, disease activity, as indicated by mean ASDAS-CRP values, remained stable throughout 2 years of observation.

Safety (safety set)

Overall, 81.5% (66 of 81) and 53.7% (58 of 108) of patients with PsA and r-axSpA, respectively, experienced at least one TEAE, 21.0% (17 of 81) and 14.8% (16 of 108) of patients, respectively, experienced at least one treatment-related TEAE, and 11.1 (9 of 81) and 13.9% (15 of 108), respectively, discontinued treatment with secukinumab because of TEAEs (Table 4). Among patients with PsA, 15 SAEs were reported, with 13.6% of patients (11 of 81) experiencing at least one; among patients with r-axSpA, nine SAEs were reported, with 8.3% of patients (9 of 108) experiencing at least one. One



Figure 2. Time to treatment discontinuation by history of treatment with biologic agents. (A) Treatment naïve. (B) Biologic pre-treated. r-axSpA: radiographic axial spondyloarthritis

Table 3. Efficacy responses with secukinumab in PsA and radiographic axial spondyloarthritis cohort up to 2 years (target population set)

Endpoints	Enrolment		Year 2	
$-\frac{1}{PsA(N=78)}$				
Swollen joint count, mean \pm s.p. (<i>M</i>)	$4.5 \pm 6.0 (36)$	3.3 ± 4.6 (27)	$3.6 \pm 4.1 (14)$	
Tender joint count, mean \pm s.D. (<i>M</i>)	12.8 ± 16.0 (36)	10.7 ± 11.3 (27)	9.2 ± 9.3 (14)	
Presence of dactylitis, n/M (%)	8/67 (11.9)	3/52 (5.8)	3/35 (8.6)	
Presence of enthesitis, n/M (%)	17/27 (63.0)	12/29 (41.4)	9/12 (75.0)	
Total pain (VAS), mean \pm s.D. (<i>M</i>)	38.9 ± 23.0 (39)	42.5 ± 26.6 (41)	56.1 ± 26.2 (18)	
PGA $0/1$ response, n/M (%)	14/24 (58.3)	19/32 (59.4)	9/11 (81.8)	
Nail involvement, n/M (%)	9/65 (13.8)	4/50 (8.0)	5/36 (13.9)	
HAQ-DI, mean \pm s.D. (M)	$1.1 \pm 0.8 (35)$	$1.1 \pm 0.7 (30)$	$1.1 \pm 0.7 (11)$	
FACIT-Fatigue, mean \pm s.d. (M)	29.8 ± 11.3 (32)	29.8 ± 12.1 (30)	$30.5 \pm 10.0 (11)$	
Total PASI score, mean \pm s.d. (<i>M</i>)	1.5 ± 4.3 (24)	1.5 ± 2.3 (21)	0.6 ± 1.3 (9)	
r-axSpA (N = 105)				
BASDAI, mean \pm s.d. (M)	3.4 ± 2.7 (91)	2.9 ± 2.7 (46)	2.3 ± 2.9 (21)	
Nocturnal back pain (VAS), mean \pm s.D. (<i>M</i>)	37.1 ± 29.4 (60)	27.6 ± 29.3 (33)	$16.3 \pm 23.0 (12)$	
Total back pain (VAS), mean \pm s.D. (M)	35.2 ± 28.2 (74)	31.9 ± 30.6 (48)	18.6 ± 26.7 (17)	
$\operatorname{CRP} > 5 n/M (\%)$	17/45 (37.8)	16/41 (39.0)	7/15 (46.7)	
ASDAS-CRP mean \pm s.D. (<i>M</i>)	2.4 ± 1.1 (28)	2.3 ± 1.1 (20)	2.4 ± 1.6 (8)	
Patient's global assessment of disease activity, mean \pm s.d. (<i>M</i>)	5.3 ± 2.6 (64)	$4.9 \pm 2.8 (36)$	4.8 ± 2.8 (16)	
FACIT-Fatigue, mean \pm s.d. (<i>M</i>)	27.3 ± 11.8 (59)	29.6 ± 9.6 (26)	30.7 ± 10.2 (10)	

ASDAS: Ankylosing Spondylitis DAS; FACIT-Fatigue: Functional Assessment of Chronic Illness Therapy—Fatigue subscale score; HAQ-DI: HAQ—Disability Index; *M*: number of patients with evaluation; *n*: number of patients; *N*: number of patients in the study population; PASI: Psoriasis Area and Severity Index; PGA: Physician's Global Assessment; r-axSpA: radiographic axial spondyloarthritis; VAS: visual analogue scale (0–100).

Table 4. Overall safety profile in PsA and r-axSpA patients (safety set)

Variable	PsA (<i>n</i> = 81)		r-axSpA (<i>n</i> = 108)	
Total exposure time, years	178.6		199.0	
Mean exposure time, years	2	.2	1.8	
Patients with at least one TEAE, n (%)	66 (81.5)		58 (53.7)	
Patients with at least one treatment-related TEAE, $n (\%)^{a}$	17 (21.0)		16 (14.8)	
SAE ^b	15		9	
Patients with at least one SAE, $n(\%)$	11 (13.6)		9 (8.3)	
Patients with TEAEs lead- ing to death, n (%)	0		1 (0.9) ^c	
Patients with TEAEs lead- ing to discontinuation, n $(\%)^{c}$	9 (11.1)		15 (13.9)	
TEAEs of special interest	n (%)	EAIR ^d	n (%)	EAIR ^d
Candida infections	1 (1.2)	0.5	1(0.9)	0.5
Oral candidiasis	0	0	1 (0.9)	0.5
Malignancy	0	0	1(0.9)	0.5
MAČE	1(1.2)	0.5	2 (1.9)	1.0
IBD	0	0	0	0
Uveitis	0	0	0	0

^a AE/SAE with relationship 'suspected'.

^b Includes infections and infestations (n = 6: *Campylobacter* gastroenteritis, diverticulitis, lower respiratory infection, tooth abscess, tuberculosis, urosepsis); cardiac disorders [n = 3: acute myocardial infection (2), atrial fibrillation]; renal and urinary disorders (n = 3: nephrolithiasis, post-streptococcal glomerulonephritis, renal failure); nervous system disorders (n = 2: ataxia, ischaemic stroke); gastrointestinal disorders (n = 2: abdominal hernia, umbilical hernia); anaemia; cholecystitis; osteonecrosis; respiratory failure; psoriasis; metastatic malignant melanoma; death.

Not related to treatment.

^d Incidence rate per 100 patient-years calculated as: [total number of events in study interval/total patient (years in study interval[†])] ×100. [†](Date of study completion or end date of study interval or date of discontinuation or interim cut-off date – date of informed consent + 1)/365.25. EAIR: exposure-adjusted incidence rate; MACE: major adverse cardiac event; TEAE: treatment-emergent adverse event.

death was reported during the observation period, which was considered unrelated to secukinumab treatment.

The overall incidence of AEs of special interest was low in patients with PsA or r-axSpA (Table 4). Overall, three patients experienced MACEs, including ischaemic stroke in a 46-year-old patient with PsA who was a current smoker at study start, and acute myocardial infarction in two patients with r-axSpA (a 45-year-old patient who was a current smoker at study start and a 48-year-old patient who had never smoked). Two patients developed *Candida* infections. One patient with PsA receiving concomitant treatment with fluticasone developed mild laryngeal candidiasis, which prompted a temporary (6-week) interruption of secukinumab therapy. One patient with r-axSpA developed mild oral candidiasis, which resolved without any changes to the secukinumab treatment regimen. One patient with r-axSpA was diagnosed with a malignancy (malignant melanoma). No new cases or flares of IBD or uveitis were reported during the study.

Discussion

A large number of international clinical trials have confirmed the safety and efficacy of secukinumab in patients with PsA or axSpA [25, 26, 27, 28, 29, 30, 41]. This interim analysis of UK data from the ongoing SERENA trial provides insight into how secukinumab is used and how patients respond in a UK health-care setting and facilitates comparison with other real-world data on secukinumab use. The National Institute for Health and Care Excellence (NICE) recommends secukinumab as an option for treating active PsA in patients with three or more swollen joints and who have not responded to at least two standard DMARDs or who have not responded to TNF- α inhibitors [42], and for treating active r-axSpA in adults whose condition has responded inadequately to conventional therapies [43]. The population included in the present study was treated in accordance with these recommendations, in that patients had generally received

extensive treatment with NSAIDs, DMARDs and biologic therapies before initiating treatment with secukinumab; some biologic-experienced patients had received three or more biologics before starting secukinumab. Furthermore, NICE recommends that the response to treatment should be assessed after 16 weeks in patients with PsA or r-axSpA and continued only if there is clear evidence of response [42, 43]. The SERENA protocol specified that patients were eligible for enrolment after 16 weeks of treatment [32], meaning that the baseline data correspond to the point at which NICE recommends evaluating the response to treatment. This means that there is a high likelihood that only patients with a satisfactory response to treatment were enrolled.

High retention rates of $\sim 91\%$ at 1 year and $\sim 77\%$ at 2 years were observed in UK patients with PsA or r-axSpA in the present analysis. These results are similar to the overall SERENA cohort (n = 1004), in which retention rates were 85% and 86%, respectively, in patients with PsA and AS at 1 year, and 75% and 79%, respectively, at 2 years [33]. There are several differences between the UK cohort and the overall population (438 sites across Europe) that might contribute to the slightly better retention rates in the UK after 1 year of treatment. The duration of secukinumab treatment before inclusion was shorter in the UK subgroup, both in patients with PsA (0.5 vs 1.0 years in the overall population) and r-axSpA (0.7 vs 0.9 years in the overall population) [33]. Thus, there is a difference in the duration of treatment with secukinumab and the duration of observation between the UK cohort and the overall population. A lower proportion of patients in the UK cohort (56% with PsA and 57% with r-axSpA vs 67% and 63%, respectively) had received prior treatment with a biologic agent [33]. Conversely, a higher proportion of patients in the UK cohort vs the overall population were affected by obesity (46% vs 26%, respectively) at baseline [33], which has been associated with higher secukinumab treatment retention in women in one small observational study [44]. In addition, a higher proportion of patients in the UK cohort vs the overall population were current or former smokers (48% vs 38%), and a higher proportion of patients with PsA had dactylitis (12% vs 6%) or enthesitis (63% vs 19%) at baseline [33], the impact of which on treatment response is uncertain. Given that the baseline data represent results after ~ 16 weeks of treatment, it is not possible to gauge the extent of improvement from the start of secukinumab treatment. However, the data show that responses obtained after ~ 16 weeks of treatment were sustained over the next 2 years of treatment with secukinumab.

To provide a context for interpreting these data, prior observational studies in the UK have reported retention rates of 79% and 73% after the first and second years of treatment with a first TNF- α inhibitor in patients with PsA [45], and 83% after a first TNF- α inhibitor in patients with axSpA [46].

Data on drug survival of secukinumab are becoming available from other real-world studies in patients with PsA or AS and show that secukinumab is effective and has a high retention rate over time [27–30]. It should be noted that these results pertain to retention from the start of secukinumab treatment and therefore differ from the present study, in which observation began after 16 weeks of treatment; thus, the reported retention rates are not comparable. Analysis of data from >2000 patients with PsA treated with secukinumab in 13 clinical practice registries participating in the European Spondyloarthritis Research Collaboration Network showed that secukinumab treatment was retained by 86% of patients after 6 months and by 76% after 12 months [28]. Higher retention rates were observed in biologic/targeted synthetic DMARD-naïve patients than in patients who had received two or more prior biologic/targeted synthetic DMARD agents at 6 months (90% vs 85%, respectively) and 12 months (82% vs 72%) [28], which shows that more extensive prior treatment is associated with somewhat lower response rates.

A nationwide cohort study (n = 2551) showed that, among Danish patients with PsA who received treatment with biologics and immunomodulators, drug survival rates were higher for agents that target IL-17, including secukinumab, alongside agents that target TNF and IL-12/23 [29]. The same analysis showed that, in patients with AS, drug survival by secukinumab was among the highest, both overall and in biologicexperienced patients in particular [29]. Likewise, among French patients with PsA (n = 6531), treatment with an agent targeting IL-17 (individual agents were not specified) was associated with higher persistence than treatment with agents targeting TNF or IL-12/23 [31].

Despite the Phase 3 EXCEED study showing greater secukinumab retention rates *vs* adalimumab when administered as first-line treatment [47], analyses of data collected in Nordic countries between 2015 and 2018 showed that there was no significant difference in treatment retention rates between adalimumab and secukinumab when used to treat patients with PsA [27] or as first- or second-line therapy for SpA [30]. That said, secukinumab was rarely used as first-line treatment [27].

Efficacy was not the primary objective of SERENA, and efficacy results are missing for many patients. However, among patients with PsA for whom efficacy data are available, the mean TJCs (4.5 ± 6.0) and SJCs (12.8 ± 16.0) at enrolment (~6 months after starting treatment with secukinumab) are relatively high. This suggests that the disease must have been severe at baseline. Although it is not possible to comment on the extent of improvement between the start of treatment with secukinumab and the time of enrolment in SERENA, these data suggest that the digree of improvement after ~6 months provided encouragement to continue treatment. In contrast, among patients with r-axSpA who provided data, the mean BASDAI score (3.4 ± 2.7) suggests that the disease was well controlled at enrolment and was in remission after 1 and 2 years.

In the SERENA UK cohort, treatment interruptions occurred infrequently and were attributed to diverse reasons, although it should be noted that AEs were not the most common reason for treatment interruptions. The most common reasons for discontinuation of secukinumab were lack of efficacy, physician decision and patient decision, whereas AEs were cited by few patients, which suggests that treatment was generally well tolerated. This finding is consistent with those of the overall SERENA study, in which the major reasons for drug discontinuation were lack of efficacy and patient decision [33]. The safety profile of secukinumab in the present study is consistent with that observed in large Phase 3 studies in patients with PsA [14-18] or r-axSpA [19, 22]. No new safety signals were detected; the incidence of TEAEs of special interest was low, and no new cases of uveitis or IBD were reported. This is consistent with analyses of data from Phase 3 clinical trials suggesting that there is a low risk of uveitis or IBD with secukinumab treatment [48, 49].

Limitations of this real-world analysis have been described in detail elsewhere [32]. The most pertinent of these include the observational study design, lack of a control group, limited or missing data, and the small number of patients in the present analysis because it was restricted to one country. Given that patients were enrolled after receiving 16 weeks of treatment with secukinumab, the study is also subject to selection bias, because only patients with an initial response to secukinumab were included. Thus, the findings apply only to patients who are eligible to continue secukinumab treatment after having completed 16 weeks of initial treatment.

In conclusion, this analysis of real-world data from the UK demonstrates high retention rates and sustained efficacy of secukinumab over >2 years in patients with PsA or r-axSpA, consistent with the overall results of SERENA and with emerging data from other real-world studies. The overall incidence of TEAEs and TEAEs of special interest was low. The safety profile is consistent with that reported in clinical trials in patients with PsA or r-axSpA, and no new safety signals were observed.

Supplementary material

Supplementary material is available at *Rheumatology Advances in Practice* online.

Data availability

Data are available upon reasonable request. All data relevant to the study are included in the article or uploaded as Supplementary Material (available at *Rheumatology Advances in Practice* online). The data sets generated and/or analysed during the current study are not publicly available. Novartis is committed to sharing with qualified external researchers' access to patient-level data and supporting clinical documents from eligible studies. These requests are reviewed and approved on the basis of scientific merit. All data provided are anonymized to respect the privacy of patients who have participated in the study in line with applicable laws and regulations. The data may be requested by writing to the corresponding author.

Funding

This work was supported by Novartis Pharma AG.

Disclosure statement: L.H. and P.P. are employees of Novartis Pharmaceuticals UK Ltd. N.G. has received research support, consultancy fees and/or speakers fees from AbbVie, Celgene, Eli Lilly, Izana, Janssen, Novartis and UCB. K.G. reports consultancy for AbbVie, Celgene, Lilly, Pfizer, Gilead, MSD, Novartis, UCB, member of speakers' bureau for AbbVie, Celgene, Lilly, Pfizer, Gilead, MSD, Novartis, UCB and grants/research support from AbbVie, Celgene, Lilly, Pfizer, Gilead, MSD, Novartis, UCB. R.S. has received honoraria from AbbVie, Biogen, Celgene, Lilly, MSD, Novartis, Roche and UCB, and grants/research support from AbbVie, Celgene, Novartis and UCB.

Acknowledgements

Medical writing support was provided by Blair Jarvis, MSc and Imogen Allred, DPhil, of OPEN Health Communications (London, UK) and funded by Novartis Pharmaceuticals UK Ltd. Some of these data were presented at a Novartis Promotional Dinner as a presentation entitled 'Updates in Rheumatology at Novartis'.

References

- 1. Taurog JD, Chhabra A, Colbert RA. Ankylosing spondylitis and axial spondyloarthritis. N Engl J Med 2016;374:2563–74.
- Ritchlin CT, Colbert RA, Gladman DD. Psoriatic arthritis. N Engl J Med 2017;376:957–70.
- 3. Fernández-Carballido C, Martín-Martínez MA, García-Gómez C *et al.*; Cardiovascular in Rheumatology Project Collaborative Group. Impact of comorbidity on physical function in patients with ankylosing spondylitis and psoriatic arthritis attending rheumatology clinics: results from a cross-sectional study. Arthritis Care Res (Hoboken) 2020;72:822–8.
- Mease PJ, Liu M, Rebello S *et al.* Disease characteristics, quality of life, and work productivity by enthesitis site: real-world data from the US Corrona Psoriatic Arthritis/Spondyloarthritis Registry. J Rheumatol 2021;48:367–75.
- 5. Gudu T, Gossec L. Quality of life in psoriatic arthritis. Expert Rev Clin Immunol 2018;14:405–17.
- Ogdie A, Langan S, Love T *et al.* Prevalence and treatment patterns of psoriatic arthritis in the UK. Rheumatology (Oxford) 2013;52: 568–75.
- Kotsis K, Voulgari PV, Drosos AA, Carvalho AF, Hyphantis T. Health-related quality of life in patients with ankylosing spondylitis: a comprehensive review. Expert Rev Pharmacoecon Outcomes Res 2014;14:857–72.
- Derakhshan MH, Dean L, Jones GT, Siebert S, Gaffney K. Predictors of extra-articular manifestations in axial spondyloarthritis and their influence on TNF-inhibitor prescribing patterns: results from the British Society for Rheumatology Biologics Register in Ankylosing Spondylitis. RMD Open 2020;6:e001206.
- 9. Hamilton L, Macgregor A, Toms A *et al*. The prevalence of axial spondyloarthritis in the UK: a cross-sectional cohort study. BMC Musculoskelet Disord 2015;16:392.
- 10. Le Goff B, Bouvard B, Lequerre T *et al.* Implication of IL-17 in bone loss and structural damage in inflammatory rheumatic diseases. Mediators Inflamm 2019;2019:8659302.
- Blauvelt A, Chiricozzi A. The immunologic role of IL-17 in psoriasis and psoriatic arthritis pathogenesis. Clin Rev Allergy Immunol 2018;55:379–90.
- 12. Cosentyx summary of product characteristics. Dublin, Ireland: Novartis Europharm Limited, 2019.
- 13. Bissonnette R, Luger T, Thaçi D *et al.* Secukinumab demonstrates high sustained efficacy and a favourable safety profile in patients with moderate-to-severe psoriasis through 5 years of treatment (SCULPTURE Extension Study). J Eur Acad Dermatol Venereol 2018;32:1507–14.
- McInnes IB, Mease PJ, Kirkham B *et al.*; FUTURE 2 Study Group. Secukinumab, a human anti-interleukin-17A monoclonal antibody, in patients with psoriatic arthritis (FUTURE 2): a randomised, doubleblind, placebo-controlled, phase 3 trial. Lancet 2015;386:1137–46.
- Mease PJ, McInnes IB, Kirkham B *et al.*; FUTURE 1 Study Group. Secukinumab inhibition of interleukin-17A in patients with psoriatic arthritis. N Engl J Med 2015;373:1329–39.
- 16. Nash P, Mease PJ, McInnes IB *et al.*; FUTURE 3 Study Group. Efficacy and safety of secukinumab administration by autoinjector in patients with psoriatic arthritis: results from a randomized, placebocontrolled trial (FUTURE 3). Arthritis Res Ther 2018;20:47.
- 17. McInnes IB, Mease PJ, Kivitz AJ *et al*. Long-term efficacy and safety of secukinumab in patients with psoriatic arthritis: 5-year (end-of-study) results from the phase 3 FUTURE 2 study. Lancet Rheumatol 2020;2:e227–35.
- 18. Mease PJ, Kavanaugh A, Reimold A *et al.*; FUTURE 1 Study Group. Secukinumab provides sustained improvements in the signs and symptoms of psoriatic arthritis: final 5-year results from the phase 3 FUTURE 1 study. ACR Open Rheumatol 2020;2:18–25.

- Pavelka K, Kivitz A, Dokoupilova E *et al.* Efficacy, safety, and tolerability of secukinumab in patients with active ankylosing spondylitis: a randomized, double-blind phase 3 study, MEASURE 3. Arthritis Res Ther 2017;19:285.
- 20. Baraliakos X, Braun J, Deodhar A *et al.* Long-term efficacy and safety of secukinumab 150 mg in ankylosing spondylitis: 5-year results from the phase III MEASURE 1 extension study. RMD Open 2019;5:e001005.
- 21. Marzo-Ortega H, Sieper J, Kivitz AJ *et al.* 5-year efficacy and safety of secukinumab in patients with ankylosing spondylitis: end-of-study results from the phase 3 MEASURE 2 trial. Lancet Rheumatol 2020;2:e339–46.
- 22. Pavelka K, Kivitz AJ, Dokoupilova E *et al.*; MEASURE 3 Study Group. Secukinumab 150/300 mg provides sustained improvements in the signs and symptoms of active ankylosing spondylitis: 3-year results from the phase 3 MEASURE 3 study. ACR Open Rheumatol 2020;2:119–27.
- 23. Deodhar A, Blanco R, Dokoupilová E *et al.* Improvement of signs and symptoms of nonradiographic axial spondyloarthritis in patients treated with secukinumab: primary results of a randomized, placebocontrolled phase III study. Arthritis Rheumatol 2021;73:110–20.
- Kiltz U, Peterlik D, Winkelmann V, Tony H; AQUILA_Study_Group. FRI0406 AQUILA study in Germany – real world data on secukinumab's effectiveness in ankylosing spondylitis patients – results from an interim analysis. Ann Rheum Dis 2019;78(Suppl 2):890.
- 25. Chimenti MS, Fonti GL, Conigliaro P *et al.* One-year effectiveness, retention rate, and safety of secukinumab in ankylosing spondylitis and psoriatic arthritis: a real-life multicenter study. Expert Opin Biol Ther 2020;20:813–21.
- 26. Yiu ZZN, Mason KJ, Hampton PJ *et al.*; BADBIR Study Group. Drug survival of adalimumab, ustekinumab and secukinumab in patients with psoriasis: a prospective cohort study from the British Association of Dermatologists Biologics and Immunomodulators Register (BADBIR). Br J Dermatol 2020;183:294–302.
- 27. Lindström U, Glintborg B, Di Giuseppe D *et al.* Comparison of treatment retention and response to secukinumab versus tumour necrosis factor inhibitors in psoriatic arthritis. Rheumatology (Oxford) 2021;60:3635–45.
- Michelsen B, Georgiadis S, Di Giuseppe D *et al.* Real-world sixand twelve-month drug retention, remission, and response rates of secukinumab in 2,017 patients with psoriatic arthritis in thirteen European countries. Arthritis Care Res (Hoboken) 2022;74: 1205–18.
- 29. Egeberg A, Rosenø NAL, Aagaard D *et al.* Drug survival of biologics and novel immunomodulators for rheumatoid arthritis, axial spondyloarthritis, psoriatic arthritis, and psoriasis - A nationwide cohort study from the DANBIO and DERMBIO registries. Semin Arthritis Rheum 2022;53:151979.
- 30. Glintborg B, Lindström U, Giuseppe DD et al.; Danish Rheumatology Database (DANBIO), Anti-Rheumatic Therapy in Sweden/Swedish Rheumatology Quality (ARTIS/SRQ), Center for Rheumatology Research (ICEBIO), Finnish Register of Biological Treatment (ROB-FIN), and Norwegian Antirheumatic Drug Register (NOR-DMARD) registries. One-year treatment outcomes of secukinumab versus tumor necrosis factor inhibitors in spondyloarthritis: results from five Nordic biologic registries including more than 10,000 treatment courses. Arthritis Care Res (Hoboken) 2022;74:748–58.
- 31. Pina Vegas L, Penso L, Sbidian E, Claudepierre P. Long-term persistence of first-line biologics for psoriatic arthritis and psoriasis: a cohort study of 23,423 patients from the French Health Insurance Database (SNDS). Abstract POS0075. Ann Rheum Dis 2022;81 (Suppl 1):255.
- 32. Kiltz U, Sfikakis PP, Gaffney K *et al.* Secukinumab use in patients with moderate to severe psoriasis, psoriatic arthritis and ankylosing spondylitis in real-world setting in Europe: baseline data from SERENA study. Adv Ther 2020;37:2865–83.

- 33. Kiltz U, Sfikakis PP, Gaffney K *et al.* Interim 2-year analysis from SERENA: a real-world study in patients with psoriatic arthritis or ankylosing spondylitis treated with secukinumab. Rheumatol Ther 2022;9:1129–42.
- 34. Gladman DD, Mease PJ, Krueger G *et al*. Outcome measures in psoriatic arthritis. J Rheumatol 2005;32:2262–9.
- 35. Mease PJ. Measures of psoriatic arthritis: Tender and Swollen Joint Assessment, Psoriasis Area and Severity Index (PASI), Nail Psoriasis Severity Index (NAPSI), Modified Nail Psoriasis Severity Index (mNAPSI), Mander/Newcastle Enthesitis Index (MEI), Leeds Enthesitis Index (LEI), Spondyloarthritis Research Consortium of Canada (SPARCC), Maastricht Ankylosing Spondylitis Enthesis Score (MASES), Leeds Dactylitis Index (LDI), Patient Global for Psoriatic Arthritis, Dermatology Life Quality Index (DLQI), Psoriatic Arthritis Quality of Life (PsAQOL), Functional Assessment of Chronic Illness Therapy-Fatigue (FACIT-F), Psoriatic Arthritis Response Criteria (PsARC), Psoriatic Arthritis Joint Activity Index (PsAJAI), Disease Activity in Psoriatic Arthritis (DAPSA), and Composite Psoriatic Disease Activity Index (CPDAI). Arthritis Care Res (Hoboken) 2011;63 Suppl 11:S64–85.
- Healy PJ, Helliwell PS. Measuring clinical enthesitis in psoriatic arthritis: assessment of existing measures and development of an instrument specific to psoriatic arthritis. Arthritis Rheum 2008;59: 686–91.
- 37. Hawker GA, Mian S, Kendzerska T, French M. Measures of adult pain: Visual Analog Scale for Pain (VAS Pain), Numeric Rating Scale for Pain (NRS Pain), McGill Pain Questionnaire (MPQ), Short-Form McGill Pain Questionnaire (SF-MPQ), Chronic Pain Grade Scale (CPGS), Short Form-36 Bodily Pain Scale (SF-36 BPS), and Measure of Intermittent and Constant Osteoarthritis Pain (ICOAP). Arthritis Care Res (Hoboken) 2011;63(Suppl 11):S240–52.
- 38. Zochling J. Measures of symptoms and disease status in ankylosing spondylitis: Ankylosing Spondylitis Disease Activity Score (ASDAS), Ankylosing Spondylitis Quality of Life Scale (ASQoL), Bath Ankylosing Spondylitis Disease Activity Index (BASDAI), Bath Ankylosing Spondylitis Functional Index (BASFI), Bath Ankylosing Spondylitis Global Score (BAS-G), Bath Ankylosing Spondylitis Metrology Index (BASMI), Dougados Functional Index (DFI), and Health Assessment Questionnaire for the Spondylarthropathies (HAQ-S). Arthritis Care Res (Hoboken) 2011;63 Suppl 11:S47–58.
- Machado PM, Landewé RB, van der Heijde DM. Endorsement of definitions of disease activity states and improvement scores for the Ankylosing Spondylitis Disease Activity Score: results from OMERACT 10. J Rheumatol 2011;38:1502–6.
- Knop C, Oeser M, Bastian L *et al.* [Development and validation of the Visual Analogue Scale (VAS) Spine Score]. Unfallchirurg 2001;104: 488–97.
- Kiltz U, Peterlik D, Winkelmann V, Tony H, AQUILA_Study_Group. FRI0405 AQUILA study in Germany – real world data on secukinumab's effectiveness in psoriatic arthritis patients – results from an interim analysis. Ann Rheum Dis 2019;78(Suppl 2):889–90.
- 42. National Institute for Health and Care Excellence. Certolizumab pegol and secukinumab for treating active psoriatic arthritis after inadequate response to DMARDs. Technology appraisal guidance [TA445] Published: 24 May 2017. https://www.nice.org.uk/guid ance/ta445 (August 2023, date last accessed).
- 43. National Institute for Health and Care Excellence. Secukinumab for active ankylosing spondylitis after treatment with non-steroidal anti-inflammatory drugs or TNF-alpha inhibitors. Technology appraisal guidance [TA407] Published: 28 September 2016. https:// www.nice.org.uk/guidance/ta407 (August 2023, date last accessed).
- 44. García-Dorta A, León-Suarez P, Peña S et al. Association of gender, diagnosis, and obesity with retention rate of secukinumab in spondyloarthropathies: results from a multicenter real-world study. Front Med 2021;8:815881.
- 45. Stober C, Ye W, Guruparan T *et al.* Prevalence and predictors of tumour necrosis factor inhibitor persistence in psoriatic arthritis. Rheumatology (Oxford) 2018;57:158–63.

- 46. Yahya F, Gaffney K, Hamilton L *et al.*; for BRITSpA. Tumour necrosis factor inhibitor survival and predictors of response in axial spondyloarthritis—findings from a United Kingdom cohort. Rheumatology (Oxford) 2018;57:619–24.
- 47. Gottlieb AB, Merola JF, Reich K *et al.* Efficacy of secukinumab and adalimumab in patients with psoriatic arthritis and concomitant moderate-to-severe plaque psoriasis: results from EXCEED, a randomized, double-blind head-to-head monotherapy study. Br J Dermatol 2021;185:1124–34.
- 48. Schreiber S, Colombel J-F, Feagan BG *et al.* Incidence rates of inflammatory bowel disease in patients with psoriasis, psoriatic arthritis and ankylosing spondylitis treated with secukinumab: a retrospective analysis of pooled data from 21 clinical trials. Annals of the Rheumatic Diseases 2019;78:473–9.
- 49. Deodhar AA, Miceli-Richard C, Baraliakos X *et al.* Incidence of uveitis in secukinumab-treated patients with ankylosing spondylitis: pooled data analysis from three phase 3 studies. ACR Open Rheumatol 2020;2:294–9.